Author's response to reviews

Title: Bladder inflammatory transcriptome in response to tachykinins: Neurokinin 1 receptor-dependent genes and transcription regulatory elements

Authors:

- Ricardo Saban (ricardo-saban@ouhsc.edu)
- Cindy Simpson (cindy-simpson@ouhsc.edu)
- Rajanikanth Vadigepalli (raj@mail.dbi.tju.edu)
- Sylvie Memet (symemet@pasteur.fr)
- Igor Dozmorov (igor-dozmorov@omrf.ouhsc.edu)
- Marcia R Saban (marcia-saban@ouhsc.edu)

Version: 4 Date: 18 May 2007

Author's response to reviews:

1-Changes were performed as requested
2-Tables will remain as additional files
3-I was not able to add to area of interest = urology (the field remains blank)